

Citation: Veletzky L, Eberhardt KA, Hergeth J, Stelzl DR, Zoleko Manego R, Mombo-Ngoma G, et al. (2022) Distinct loiasis infection states and associated clinical and hematological manifestations in patients from Gabon. PLoS Negl Trop Dis 16(9): e0010793. https://doi.org/10.1371/ journal.pntd.0010793

Editor: Sabine Specht, University of Zurich, SWITZERLAND

Received: May 23, 2022

Accepted: September 6, 2022

Published: September 19, 2022

Copyright: © 2022 Veletzky et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: This work was supported by the Federal Ministry of Science, Research and Economy of Austria as part of the European and Developing Countries Clinical Trials Partnership 2 programme, which is supported by the European Union (Grant A12/B-2 to MR) and the German Center for Infection research (TI 07.001 to LV). The funders RESEARCH ARTICLE

Distinct loiasis infection states and associated clinical and hematological manifestations in patients from Gabon

Luzia Veletzky ^{1,2,3,4}*, Kirsten Alexandra Eberhardt², Jennifer Hergeth³, Daniel Robert Stelzl^{3,5}, Rella Zoleko Manego^{2,3,6}, Ghyslain Mombo-Ngoma^{2,3,6}, Ruth Kreuzmair³, Gerrit Burger^{3,6}, Ayôla Akim Adegnika^{3,6}, Selidji Todagbe Agnandji^{3,6}, Pierre Blaise Matsiegui⁷, Michel Boussinesq⁸, Benjamin Mordmüller^{6,9}, Michael Ramharter^{2,3,4}*

1 Department of Medicine I, Division of Infectious Diseases and Tropical Medicine, Medical University of Vienna, Vienna, Austria, 2 Department of Tropical Medicine, Bernhard Nocht Institute for Tropical Medicine & I. Dep. of Medicine, University Medical Center Hamburg-Eppendorf, Hamburg, Germany, 3 Centre de Recherches Médicales de Lambaréné, Lambaréné, Gabon, 4 German Center For Infection Research (DZIF), Hamburg-Borstel-Riems, Germany, 5 Department of Urology, University Medical Center Hamburg-Eppendorf, Hamburg, Germany, 6 Institute of Tropical Medicine, University of Tübingen, Tübingen, Germany & German Center for Infection Research, partner site Tübingen, Tübingen, Germany, 7 Centre de Recherches Médicales de la Ngounié, Fougamou, Gabon, 8 Institut de Recherche pour le Développement (IRD), UMI 233-TransVIHMI-Inserm U1175-University of Montpellier, Montpellier, France, 9 Radboud University Medical Center, Department of Medical Microbiology, Nijmegen, The Netherlands

* luzia.veletzky@meduniwien.ac.at(LV); ramharter@bnitm.de (MR)

Abstract

Background

Loiasis–a filarial disease endemic in Central and West Africa–is increasingly recognized as significant individual and public health concern. While the understanding of the disease characteristics remains limited, significant morbidity and excess mortality have been demonstrated. Here, we characterize clinical and hematological findings in a large cohort from Gabon.

Methods

Loiasis-related clinical manifestations and microfilaremia, hemoglobin and differential blood counts were recorded prospectively during a cross-sectional survey. For analysis, participants were categorized into distinct infection states by the diagnostic criteria of eye worm history and microfilaremia.

Results

Analysis of data from 1,232 individuals showed that occurrence of clinical and hematological findings differed significantly between the infection states. Eye worm positivity was associated with a wide range of clinical manifestations while microfilaremia by itself was not. *Loa loa* infection was associated with presence of eosinophilia and absolute eosinophil counts were associated with extent of microfilaremia (p-adj. = 0.012, β-estimate:0.17[0.04–0.31]).

had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Conclusions

Loiasis is a complex disease, causing different disease manifestations in patients from endemic regions. The consequences for the affected individuals or populations as well as the pathophysiological consequences of correlating eosinophilia are largely unknown. Highquality research on loiasis should be fostered to improve patient care and understanding of the disease.

Author summary

Loiasis is a parasitic disease, endemic in parts of Western and Central Africa. While the disease has traditionally been considered to be benign, only recently significant disease morbidity and mortality have been shown. Most of the knowledge about loiasis, however, stems from reports on returning travelers, while comprehensive data from patients living in endemic areas are missing. Blood microfilaremia and reported eye worm are important diagnostic manifestations of the disease, but they can occur independent of each other in affected individuals. We analyzed hematological and clinical findings according to loiasis infection states, comprising reported eye worm and microfilaremia positivity, in a large group of individuals from a highly endemic area.

While we found that all loiasis infection states were significantly associated with absolute blood eosinophilia, the eosinophilia was more pronounced in microfilaremic loiasis. Further, there was an association between the extent of microfilaremia and absolute eosinophilia. Analyzing the frequency of clinical disease manifestations, we found that eye worm positive loiasis was associated with a range of symptoms, but microfilaremic loiasis was not. Summarizing, even in highly endemic populations different loiasis infection states are associated with distinct disease manifestations, underlining that loiasis is a versatile and indeed relevant disease.

Introduction

Loiasis, caused by the filaria *Loa loa*, is a parasitic disease endemic in rural West and Central Africa [1]. It is transmitted by tabanid flies of the genus *Chrysops* and 14 million people are estimated to be infected. In hyperendemic regions prevalence often exceeds 60% in adults [2]. Loiasis is commonly known for the pathognomonic sign of the "eye worm", which is caused by adult *L. loa* filariae transiently migrating under the patient's conjunctiva. This distinct clinical feature has been used as an epidemiological marker to assess community prevalence of loiasis in endemic settings ("RAPLOA" survey) [2].

Adult *L. loa* filariae can live more than 15 years in their definitive hosts, continuously migrating through the soft tissue and reproducing sexually. Female worms release microfilariae (mf) which can be found in the peripheral blood of patients, in sometimes very high numbers (up to 300,000 mf/mL) [1,3]. Loiasis without detectable microfilaremia is commonly referred to as "occult loiasis". Determinants for the presence or level of microfilaremia are incompletely understood and may include host genetics, varying immune responses, parasite factors such as number of adult worms, worm fecundity or sex ratio [4–6]. The majority of clinical data in scientific literature stem from reports on travelers and temporary residents of endemic regions [1,7–11]. Only few studies systematically evaluated the clinical picture of loiasis in long-term residents of endemic regions, not least due to the supposedly benign disease

course [4,12–17]. Chronic infection has been associated with unspecific manifestations, such as itching, body pains or fatigue and importantly, individuals with high microfilarial densities are known to be at risk for spontaneous or treatment-related severe manifestations and excess mortality [1,7,12,13,18,19]. Eosinophils are known to be important immune-effector cells in parasitic infections and an association between loiasis and eosinophilia has been described in short term residents, but their role in chronic loiasis has barely been studied [8,9,12].

While loiasis is in general increasingly understood as an infection of clinical relevance, many questions on the clinical features and manifestations in the most affected populations are still unanswered. To improve our understanding of the disease, we evaluated clinical and hematological findings associated with loiasis in a large cohort of individuals living in rural Gabon [18].

Methods

Ethics statement

The study was conducted in accordance with the Declaration of Helsinki, ICH-GCP guidelines and local regulations. Ethical approval was obtained from the Comité d'Ethique Institutionnel du Centre de Recherches Médicales de Lambaréné; IORG0007336/IRB00008812. Project aims and procedures were presented to local authorities, who agreed to all study-related activities. All individuals above one year of age were invited to participate and study-related procedures were only initiated after provision of written informed consent by participants or their legal representatives.

Data collection

Detailed methods of data collection were described previously [18]. In brief, data were gathered in a cross-sectional survey conducted in 2017–2018 in a rural region of Gabon where *L. loa* is endemic [3,20]. A standardized questionnaire was used to collect baseline demographic data and loiasis-related complaints, including the RAPLOA questionnaire to inform on life-time and within the last year eye worm history [2]. 5 mL of peripheral blood were collected from all participants between 10 am and 3 pm in an EDTA tube. Samples were handled on the same day following standardized procedures.

Diagnostics

Microfilaria diagnostics included thick blood smears and concentration technique which have been described in detail before [18]. Two Giemsa-stained thick blood smears of 10 μ L each were examined under a microscope and, if no mf were found, additional examination of 1 mL whole blood was performed after lysis and centrifugation. *L. loa* and *M. perstans* mf were distinguished by trained microscopists based on size, head and tail characteristics [21]. Hemoglobin levels and white blood cell (WBC) counts were measured using rapid diagnostic devices (HemoCue WBC System, HemoCue Hb 201+) or Pentra hematology analyzers. Results of both machines were compared in a subgroup and shown to be comparable.

M. perstans and *M. sp. Deux* are co-endemic in the study area. As microfilarial densities are often low in mansonellosis, DNA was extracted from 500 μ l whole blood and analyzed by polymerase chain reaction (PCR) to increase diagnostic sensitivity [3]. Detailed methods are provided in the supporting information.

Classified infection state	Diagnostic criterion	Abbreviation
Eye worm positive loiasis	Eye worm positivity based on the standardized RAPLOA questionnaire [2,42]	EW
Microfilaremic loiasis	Detectable midday microfilaremia assessed by microscopy of two Giemsa-stained thick blood smears (total of 20µl full blood) AND 1mL of full blood assessed after hemolysis using saponin followed by centrifugation	MF
Eye worm positive as well as microfilaremic loiasis	Individuals fulfilling both criteria	EWMF
Microfilaremic loiasis with low microfilaremia	Microfilaremia between 1–7,999 mf/mL	LMF
Microfilaremic loiasis with high microfilaremia	Microfilaremia between 8,000–19,999 mf/mL	HMF
Microfilaremic loiasis with hyper microfilaremia	Microfilaremia above ≥20,000 mf/mL	HYMF
No sign of loiasis infection	Individuals who have no history of eye worm and no detectable microfilaremia	LN

Table 1. Definitions of the classified infection states described in the paper.

https://doi.org/10.1371/journal.pntd.0010793.t001

Statistical analysis

Statistical analysis was performed using the software package R (Version 4.0.5, R Foundation for Statistical Computing, Vienna, Austria). Participants were categorized into "infection states", defined by presence/absence of eye worm history and presence/absence of detectable microfilaremia. Thus, the main *L. loa* infection states were: 1) No sign of *L. loa* infection, i.e. absence of microfilaremia and no history of eye worm (LN), 2) detectable microfilaremia but no history of eye worm (MF), 3) positive history of eye worm but no detectable microfilarei (EW) and 4) history of eye worm and detectable microfilaremia (EWMF). To assess differences according to the level of *L. loa* microfilaremia, microfilaremic participants were categorized into three groups: low (1–7,999 mf/mL; LMF), high (8,000–19,999 mf/mL; HMF) and hyper-microfilaremia (\geq 20,000 mf/mL; HYMF) (see Table 1). Occurrence of reported symptoms during the previous three months and hematological parameters were analyzed and compared between infection states. Calabar swelling was not considered as diagnostic criterion as it has been shown to be of limited specificity but was included as symptom in data analysis if it occurred during the previous year [2].

Categorical variables were compared using the χ^2 test or the Fisher exact test as appropriate. Multi-group comparisons were performed and corrected for multiple comparisons using the false discovery rate method. Continuous variables were expressed as median and interquartile range (IQR). To correct for demographic parameters and filarial co-infections, multiple regression models were run and adjusted for the following possible confounders: age group, sex and presence of *M. perstans* microfilariae. In case of logistic regression analysis, adjusted prevalence ratios (aPR, 95% confidence interval, CI) were computed.

Results

Study population

Data on history of eye worm migration and microfilaremia were available for 1,232 participants. First, the cohort was categorized into the above described four mutually exclusive infection states LN, MF, EW and EWMF. MF, EW and EWMF are thus the subgroups of individuals with definite signs for loiasis (LP).

Within the loiasis positive infection states (LP, n = 626), EW was the most common infection state (52.4%, n = 328), followed by EWMF (30.7%, n = 192) and MF (16.9%, n = 106) (see Table 2). Inter-group comparisons between the infection states revealed significant differences between age and sex distribution. There was no difference in the distribution of extent of

Table 2. Distribution of subjects according to the infection state within each age and sex category and comparisons of these distributions between infection state categories. Distributions by age categories were compared using the Fisher's exact test, and distributions by sex were compared using the χ^2 test.

nge und sex	aistii	button	Weight of fouries infection states e LN^* MF^* EW^* $EW F^*$ p -value Post hoc test (False discovery rate correction) %* N % n % n % overall Inter-group comparisons 00 606 49.2 106 8.6 328 26.6 192 15.6 MF vs. LN EW vs. LN EWMFvs. LN EW vs. MF EWMFvs. MF EW vs. EWMF 2.7 134 85.3 6 3.8 15 9.6 2 1.3														
	Whole cohort		LN*		MF*		EW*		EWMF*		p-value	Post hoc test (False discovery rate correction)					
	N	%°	N	%	n	%	N	%	n	%	overall	Inter-group comparisons					
Total	1232	100	606	49.2	106	8.6	328	26.6	192	15.6		MF vs. LN	EW vs. LN	EWMFvs. LN	EW vs. MF	EWMFvs. MF	EW vs. EWMF
Age (years)																	
<15	157	12.7	134	85.3	6	3.8	15	9.6	2	1.3							
15-59	751	61.0	364	48.5	57	7.6	212	28.2	118	15.7	< 0.001	< 0.001	< 0.001	< 0.001	0.125	0.055	0.048
≥ 60	324	26.3	108	33.3	43	13.3	101	31.2	72	22.2							
Sex																	
Female	668	54.2	343	51.4	31	4.6	209	31.3	85	12.7	< 0.001	< 0.001	0.041	0.006	< 0.001	0.019	< 0.001
Male	564	45.8	263	46.6	75	13.3	119	21.1	107	19.0							

Age and sex distributions of loiasis infection states

* LN = No sign of loiasis infection; MF = detectable microfilaremia but no history of eye worm; EW = positive history of eye worm but no detectable microfilarea; EWMF = positive history of eye worm as well as detectable microfilaremia, % = column percentages, % = row percentages

https://doi.org/10.1371/journal.pntd.0010793.t002

microfilaremia between women and men, nor between the adult age groups when comparing the microfilaremic categories (see Table 3).

Loiasis infection states and clinical presentation

Occurrence of the disease manifestations were compared between the infection states and results were adjusted to possible confounding factors including sex, age and *M. perstans* microfilaremia (see Tables A, B, C in <u>S1 Appendix</u>). Fig 1 displays the frequency of reported manifestations by infection state, and adjusted p-values of intergroup comparisons if significant. In comparison to LN, each of the two groups with history of eye worm (EW and EWMF) reported several signs and symptoms significantly more often. This was the case for Calabar swelling-like manifestations (adjusted prevalence ratio [95% confidence interval]: (1.40 [1.10–1.79] for EW and 2.05 [1.56–2.69] for EWMF), paresthesia (1.51 [1.19–1.92] and 1.73 [1.32–2.26]), transient paralysis of extremities (2.63 [1.86–3.72] and 1.78 [1.15–2.76]), and fatigue (1.81 [1.43–2.30] and 1.49 [1.12–2.01]). In addition, individuals in the EW category reported

Table 3. Distribution of subjects according to their microfilaremia category within each age and sex category and comparisons of these distributions between microfilaremia categories. Distributions by age and sex categories were compared using the Fisher's exact test.

Age and sex distributions of subgroups of microfilaremic individuals												
	N 298	LN	∕IF [∗]	H	MF*	HYMF*		p-value	Post hoc test (False discovery rate correction)			
		n	% 81.2	n 43	% 14.4	n 13	% 4.4	overall	Inter-group comparisons			
Total		242							LMF vs. HMF	LMF vs. HYMF	HMF vs. HYMF	
Age (years)												
<15	8	8	100	0	0	0	0					
15-59	175	140	80.0	28	16.0	7	4.0	0.768	0.795	0.852	0.795	
≥60	115	94	81.7	15	13.0	6	5.2					
Sex												
Female	116	96	82.8	16	13.8	4	3.5	0.808	0.866	0.866	0.866	
Male	182	146	80.2	27	14.8	9	5.0					

https://doi.org/10.1371/journal.pntd.0010793.t003



Fig 1. Frequencies of signs and symptoms within the three loiasis infection states and in uninfected individuals as bar plots and adjusted p-values of inter group comparisons, if significant. All intergroup comparisons are adjusted to sex, age and *Mansonella* PCR. * p-adj. <0.05; **p-adj. <0.01; ** p-adj. <0.001; * LN = No sign of loiasis infection; MF = detectable microfilaremia but no history of eye worm; EW = positive history of eye worm but no detectable microfilaremia.

https://doi.org/10.1371/journal.pntd.0010793.g001

myalgia and severe headache more often compared to LN (1.36 [1.01–1.82] and 1.33 [1.15–1.54], respectively). The frequency of the reported manifestations was similar between MF and LN subjects.

Comparing the 3 LP categories among each other, EW and EWMF subjects reported some manifestations more frequently than MF individuals. This was the case for myalgia (1.79 [1.08–2.97] and 1.76 [1.04–2.99], respectively) and Calabar swelling-like manifestations (1.55 [1.00–2.40] and 2.27 [1.45–3.55]). Further, EW subjects reported several disease manifestations more often than the MF subjects; this was the case for transient swellings of any body part (other swellings 1.81 [1.05–3.10]), transient paralysis of extremities (2.84 [1.48–5.47]), urticaria (2.38 [1.23–4.59]) and fatigue (1.53 [1.02–2.30]). Comparing the EW and EWMF categories, patients in the EWMF group reported significantly fewer Calabar swelling-like manifestations (0.68 [0.51–0.91]), whereas patients in the EW category indicated more often urticaria (p = 0.062, 1.48 [0.97–2.24]). HMF and HYMF individuals reported more often an episode of eye worm migration during the previous year compared to LMF (58.2% versus 35.5%; p-adj. = 0.003).

In summary, the infection states comprising participants with history of eye worm were significantly associated with the occurrence of clinical complaints. Contrarily, microfilaremic individuals reported complaints in a similar proportion as persons without signs of loiasis infection. The proportion of reported manifestations was independent of extent of microfilaremia, with exception for occurrence of eye worm.

Loiasis infection states and hematological phenotypes

Hematological findings were analyzed following the same categorization and adjustments as described above (see Table D and E in <u>S1 Appendix</u>). Data for differential blood count were



Fig 2. Distribution of hematological findings including median and interquartile range of hemoglobin (A), total white blood cell count (B), absolute (C) and relative eosinophil counts (D) within the four infection states. Adjusted p-values of intergroup comparisons are provided if <0.05. All intergroup comparisons are adjusted to sex, age and positivity of *Mansonella* PCR. * p-adj. <0.05; **p-adj. <0.01; *** p-adj. <0.001; LN = No sign of loiasis infection; MF = detectable microfilaremia but no history of eye worm; EW = positive history of eye worm but no detectable microfilaria; EWMF = positive history of eye worm as well as detectable microfilaremia.

https://doi.org/10.1371/journal.pntd.0010793.g002

incomplete or missing for 133 (10.8%) participants. There was no statistical difference in hemoglobin values between LP and LN nor between the different loiasis infection states. However, LP had higher total WBC counts (median [inter quartile range]: 7.4 [5.9–8.9] vs. 6.7 [5.4–8.4] x10³ leucocytes/ μ L, p-adj.<0.001), as well as higher absolute eosinophils/ μ L (1.1 [0.5–1.8] vs. 0.6 [0.25–1.3] x10³, p-adj.<0.001) and relative eosinophil counts (15.0% [8.0–22.0] vs. 9.2% [4.1–17.0] of total WBCC, p-adj.<0.001) compared to LN. Differences in WBC counts and eosinophilia were also found when comparing the loiasis infection states individually to LN and among each other. Fig 2 displays the results graphically including adjusted p-values.

Comparing all individuals with microfilaremia (MF and EWMF) to patients with history of eye worm only (EW), the two former categories were characterized by higher absolute eosino-phil (1.3 [0.8–1.85] and 1.4 [0.8–2.0] vs. 0.8 [0.34–1.6] $10^3/\mu$ L; p-adj. = 0.076 and 0.003) and relative eosinophil counts (18.0% [11.0–23.0] and 17.0% [12.0–24.0] vs. 13.0% [6.0–21.0]; p-adj. = 0.016 and 0.002). To further investigate this association between microfilaremia and eosinophilia, comparisons were made between the microfilaremic groups. The HYMF group

had higher absolute eosinophil counts than HMF and LMF (1.70 [0.60–2.70] vs.1.40 [0.90– 1.90] and 1.30 [0.80–1.90] $10^3/\mu$ L; p-adj. = 0.034 and p-adj. = 0.021), respectively (see Fig A in S1 Appendix). Of note, of the 56 individuals in the HMF and HYMF groups, none had an absolute eosinophil count within normal range ($\leq 0.5 \times 10^3/\mu$ L). Next, a simple linear regression analysis of the microfilaremic sub-population of the study, with at least 2 microfilariae per mL revealed an increase in absolute eosinophil count with every 10-fold increase in parasitemia (p-adj. = 0.012, ß-estimate: 0.17 [0.04–0.31], see Fig B in S1 Appendix).

In summary, loiasis was associated with absolute and relative eosinophilia. Extent of eosinophilia was higher in microfilaremic individuals in comparison to only eye worm positives. The level of eosinophilia was associated with the quantity of *L. loa* microfilaremia.

Discussion

Here we describe the clinical and hematological findings associated with *L. loa* infection in a large population of individuals living in a highly endemic region. We found that distinct infection states of loiasis were associated with specific clinical and hematological findings (see Fig <u>3A</u>). The loiasis positive infection states can be classified into three mutually exclusive categories based on diagnostic criteria. These categories consist of individuals having a positive history of eye worm (EW), microfilaremic individuals (MF) and individuals having both (EWMF) (Fig <u>3B</u>). The associated clinical and hematological outcomes do overlap in individuals who had both diagnostic criteria but seem to be associated with the specific infection states (see Fig <u>3C</u> and <u>3D</u>).

Up to now clinical and hematological findings associated with loiasis have been described predominantly in travelers and temporary residents of endemic regions. However, differences in the clinical and immunological outcomes between short term residents and individuals living in endemic regions, as in other parasitic infections, are evident [1,3,4,7–13,15]. While it is known that loiasis causes various manifestations, including signs and symptoms of a rather "allergic type" and varying degrees of immune-activation, these effects have been described to be more predominant in travelers and short term residents than in residents of endemic regions with chronic infections [10,11]. However, knowledge on disease outcomes in individuals living in endemic regions is scarce and improved understanding of the disease in this patient population is now increasingly warranted [8–10,19,22–26].

In the here presented comprehensive dataset, participants of both sexes were found to be similarly often infected with *L. loa*, however microfilaremic loiasis was found disproportionally more often in men and only eye worm positive loiasis in women. Within the adult population the occurrence and distribution of microfilaremia remained rather stable in the different age groups, a pattern that has repeatedly been described and seems to be consistent across various endemic areas [1,5,27].

Comparing the loiasis infection states between each other showed that history of eye worm migration was associated with the presence of a wide range of signs and symptoms, independently of presence of microfilaremia. At the same time, individuals reporting only history of eye worm had only moderate elevation of absolute eosinophil counts. Consequently, individuals having a high symptom load, often lack diagnostic features of infection in the blood which would facilitate definitive diagnosis, such as microfilaremia or extensive eosinophilia. This constellation may hamper diagnosis, and could cause inappropriate clinical management and treatment while these patients are indeed in need of appropriate clinical care to improve their symptoms. Additionally, these findings underline that the omission of patients reporting only history of eye worm does not only significantly underestimate the true disease prevalence, but also the extent of associated symptoms, and thus the perceived and calculated associated



Fig 3. Graphic presentation of the hematological and clinical findings within the study population. 3A is a diagram depicting the distribution and overlap of the diagnostic findings of history of eye worm, microfilaremia and eosinophilia in the study population (Eosinophilia data missing for 133 (10.8%) participants, see <u>supporting</u> information Table D). 3B depicts the distribution of the three mutually exclusive infection states in the same diagram based on diagnostic criteria. These include microfilaremic (MF), positive history of eye worm (EW), and individuals positive for both (EWMF). 3C displays the clinical outcome of eye worm positive loiasis and 3D the hematological outcome of microfilaremic loiasis in this schematic display. Note that these outcomes are not mutually exclusive, but overlap in individuals who are positive for both diagnostic criteria.

https://doi.org/10.1371/journal.pntd.0010793.g003

disease burden [1,4,14,18,24]. Importantly, we found that neurological symptoms, including paresthesia and transient paralysis of extremities, were strongly associated with loiasis, specifically with eye worm positive loiasis and not with microfilaremia itself. While these symptoms have been described in reports on returning travelers, data on their occurrence in individuals living in endemic regions are scarce [7,28,29]. However, in the here described population these neurologic findings were quite frequent in the loiasis positive individuals and should thus be further investigated in future clinical studies. In addition to the neurological symptoms, a range of unspecific symptoms, such as fatigue or headache, were quite common and should therefore be taken into account by clinicians as diagnostic hints for loiasis.

Contrarily, microfilaremic individuals reported signs and symptoms in a similar frequency as individuals with no sign of loiasis infection, indicating that microfilaremia was not in itself associated with the queried symptoms. A potential explanation for this unapparent disease presentation could be a more Th-2 driven immune response leading to a lack of "reactive" symptoms in these individuals. A small study comparing microfilaremic to amicrofilaremic individuals provided data supporting this hypothesis, but clearly, this theory still needs to be investigated in future studies [22].

Interestingly, in more historic descriptions of loiasis, when only microfilaremic individuals where diagnosed and eye worm positive loiasis was not taken into account, the infection was described to be rather asymptomatic—an observation that still shapes the perception of this disease today [7,18,19,24]. The here described finding, that microfilaremia per se is not associated with symptoms, could thus serve as an explanation for this discrepancy between current and historical disease descriptions.

Eosinophils are paramount for host defense against helminth infections and significant eosinophilia can be caused by various helminthic pathogens [30,31]. In loiasis, differences in the extent of eosinophilia and eosinophil activation have been noted between temporary residents and residents of endemic regions [8–11]. While *L. loa*-associated eosinophilia has been described in smaller cohorts, to our knowledge this is the first analysis of eosinophilia by distinct loiasis infection states in a large group of individuals from an endemic region [13]. Here, all loiasis positive infection states were associated with higher absolute eosinophil counts compared to individuals without signs of loiasis. Importantly, eosinophilia was higher in microfilaremic individuals than in amicrofilaremic loiasis. Furthermore, the extent of eosinophilia seemed to be associated with *L. loa* microfilarial density and, conclusively all 56 individuals with \geq 8,000mf/ml, had eosinophil counts above the normal range (\leq 0.5x10³/µL). The difference of absolute eosinophil counts between individuals with and without signs of loiasis and between microfilaremic and amicrofilaremic loiasis patients was statistically significant and may also be of clinical relevance. Of note, microfilaremia with related blood eosinophilia was not associated with subjective symptoms.

It is established that chronic eosinophilia can cause organ damage due to chronic inflammation [32]. The here described finding of an association between extent of eosinophilia and microfilaria counts, might therefore pathophysiologically explain the previously described excess mortality in highly microfilaremic individuals [13,19]. However, this theory needs further investigation to be verified or refuted.

Further, one could hypothesize that combining the subjective well-being of hyper-microfilaremic individuals and their previously described risk for excess mortality, could lead to inappropriately low health care seeking behavior in those most at risk for severe disease outcomes.

An important limitation of our analysis is the potential confounding of eosinophilia by other parasitic infections prevalent in the study region [3,15,20]. But, despite this co-endemicity of various parasites, hematological changes were consistently found to be associated with different infection states of loiasis, supporting the hypothesis that eosinophilia is indeed primarily caused or at least significantly augmented by *L. loa* infection. In line with this data, is the fact that the occurrence and extent of eosinophilia has long been known to depend on infection states in other filarial diseases, such as lymphatic filariasis and onchocerciasis, further supporting the here presented findings [33–35].

To understand the relevance of eosinophilia in loiasis in more detail, more clinical and immunological data are needed, especially as it is known that the provoked pathology depends on the activation of these effector cells [36]. Different immune activation mechanisms may underlie the different disease manifestations. Further longitudinal studies investigating the

association of specific immune responses, including eosinophilic activation markers with infection states, would therefore be of interest.

Importantly, the cross-sectional study design does not allow a direct inference of a causeeffect relationship between associated factors, nor the analysis of natural evolution of single versus repeated infections or assessments of more clinically serious outcomes such as organ lesions, organ damage or death [37]. Of note, due to the cross-sectional study design, laboratory findings collected at a single time point, including microfilaremia and eosinophilia, and retrospectively reported findings, including history of eye worm and occurrence of symptoms, were compared. This is a clear limitation of this study. Therefore, it is evident that the reported findings and raised hypotheses need to be addressed in prospective, clinical studies for further verification. However, due to the long duration of infections and the rare appearance of worms in the eye—a major diagnostic criterion—longitudinal studies of loiasis are extremely challenging.

In general, there are several pitfalls in loiasis diagnosis and associated difficulties have repeatedly been discussed [38]. Even in individuals where no microfilariae can be found, it is important to state that microfilaremia might still be present but just below the limit of detection of the used method. Here we used a combination of microscopy of thick blood smears and an additional concentration step assessing 1 mL of blood in order to raise the sensitivity of microfilariae detection. A qPCR based on previously published methods by Fink et al., 2011 was performed on all samples, but did not improve sensitivity and therefore results were not used for classification in this analysis (manuscript in preparation) [39]. However, very low microfilaremic individuals might have been missed and thus might have been wrongly classified as individuals without microfilaremia.

Serology has been shown to be cross-reactive and seroprevalence is very high in the study population as shown by a recent analysis (manuscript under preparation). Thus, currently available serological methods seem inappropriate to definitively distinguish active from past infection and counting all seropositive individuals as loiasis positive would likely overestimate the true disease prevalence [38,40,41]. In the absence of an adequate diagnostic tool, proving or refuting an active infection in amicrofilaremic individuals is difficult, as adult filaria might not always become evident as eye worm. In fact, it is not known which percentage of actively infected individuals never experience an eye worm migration. On the other hand, it is not known how long after a reported eye worm migration an individual remains actively infected. While it would be possible that the parasite died shortly after appearing as an eye worm, based on clinical reports it is estimated that adult L. loa filaria can live more than 15 years in the host [23]. Combining these factors, it becomes evident that history of eye worm is an imprecise diagnostic feature and does not allow to differentiate active from past infection. This may have led to misclassifications of individuals into incorrect infection states in the presented analysis. Nevertheless, the standardized questionnaire for history of eye worm (RAPLOA) remains one of the most important diagnostic tools for loiasis until better methods for detection of occult loiasis are developed. Therefore, while not allowing to exclude an infection, RAPLOA was used together with microscopy for microfilaremia, for diagnosis of loiasis in this study [2]. In conclusion, while there are obvious limitations of the data due to the cross-sectional study design, our findings are based on the best available diagnostic methods.

Further, the here provided findings were described in a large group of individuals who are likely to be re-exposed and chronically infected over several years. This underlines the significant clinical and hematological impact of loiasis, and refute once more the notion of loiasis being an inconspicuous, asymptomatic infection [1,7,12–14,19,23,37]. Contrarily, loiasis apparently shares important similarities to other filarial diseases, for which the respective infection states leading to various disease outcomes have long been acknowledged [34,35].

These data show that loiasis is a complex disease, manifesting with distinct clinical and hematological outcomes in individuals living in a highly endemic region. The consequences of these findings for the affected individuals or on populations as well as the pathophysiological consequences of correlating eosinophilia are largely unknown. Currently, loiasis remains highly neglected and individuals in endemic regions are often left untreated [18,19,24,25]. High-quality research on loiasis needs to be further fostered to provide adequate treatment options for the most affected populations and to drive the development of adequate elimination strategies.

Supporting information

S1 Appendix. Table A: Proportion of reported manifestations according to infection states. Table B: Intergroup comparisons of the proportion of reported disease manifestations in respective infection states. Table C: Intergroup comparisons of the proportion of reported disease manifestations in microfilaria density groups. Table D: Overview on hemoglobin levels and differential blood count according to different infection states of loiasis. Table E: Intergroup comparisons of hemoglobin levels and differential blood count according to differential blood count according to different infection states of loiasis. Supporting information PCR Methods. Fig A: Boxplot displaying the eosinophilia distribution by microfilaremia group (* indicates a p-adj.<0.01). Fig B: Regression plot showing the association between microfilaremia and eosinophilia. (DOCX)

S2 Appendix. This file contains the data. (XLSX)

Acknowledgments

We want to thank all communities that welcomed us for this survey and sincerely thank all volunteers for participation. Further we would like to thank all colleagues who helped during the survey, especially Madame Augustine Epee and Roger Brothier Mouckoga. We thank Nina Hackbarth for establishing and performing the *Mansonella perstans* PCR and Matthew B.B. McCall for substantial support and help to establish diagnostic procedures.

Author Contributions

- **Conceptualization:** Luzia Veletzky, Kirsten Alexandra Eberhardt, Rella Zoleko Manego, Ghyslain Mombo-Ngoma, Ayôla Akim Adegnika, Selidji Todagbe Agnandji, Michel Boussinesq, Benjamin Mordmüller, Michael Ramharter.
- **Data curation:** Luzia Veletzky, Kirsten Alexandra Eberhardt, Jennifer Hergeth, Daniel Robert Stelzl, Ruth Kreuzmair, Benjamin Mordmüller.
- **Formal analysis:** Luzia Veletzky, Kirsten Alexandra Eberhardt, Daniel Robert Stelzl, Benjamin Mordmüller.
- **Investigation:** Luzia Veletzky, Kirsten Alexandra Eberhardt, Jennifer Hergeth, Daniel Robert Stelzl, Rella Zoleko Manego, Ghyslain Mombo-Ngoma, Ruth Kreuzmair, Gerrit Burger, Ayôla Akim Adegnika, Selidji Todagbe Agnandji, Pierre Blaise Matsiegui, Benjamin Mordmüller, Michael Ramharter.
- **Methodology:** Luzia Veletzky, Ghyslain Mombo-Ngoma, Pierre Blaise Matsiegui, Benjamin Mordmüller, Michael Ramharter.

- **Project administration:** Luzia Veletzky, Jennifer Hergeth, Ruth Kreuzmair, Gerrit Burger, Selidji Todagbe Agnandji, Pierre Blaise Matsiegui, Michael Ramharter.
- **Resources:** Ghyslain Mombo-Ngoma, Ayôla Akim Adegnika, Selidji Todagbe Agnandji, Pierre Blaise Matsiegui, Benjamin Mordmüller, Michael Ramharter.

Software: Kirsten Alexandra Eberhardt.

- **Supervision:** Rella Zoleko Manego, Ghyslain Mombo-Ngoma, Ayôla Akim Adegnika, Pierre Blaise Matsiegui, Michel Boussinesq, Benjamin Mordmüller, Michael Ramharter.
- Validation: Luzia Veletzky, Jennifer Hergeth, Rella Zoleko Manego, Ghyslain Mombo-Ngoma, Ruth Kreuzmair, Michel Boussinesq, Benjamin Mordmüller, Michael Ramharter.

Visualization: Luzia Veletzky, Jennifer Hergeth.

- Writing original draft: Luzia Veletzky, Benjamin Mordmüller, Michael Ramharter.
- Writing review & editing: Luzia Veletzky, Kirsten Alexandra Eberhardt, Jennifer Hergeth, Daniel Robert Stelzl, Rella Zoleko Manego, Ghyslain Mombo-Ngoma, Ruth Kreuzmair, Gerrit Burger, Ayôla Akim Adegnika, Selidji Todagbe Agnandji, Pierre Blaise Matsiegui, Michel Boussinesq, Benjamin Mordmüller, Michael Ramharter.

References

- Boussinesq M. Loiasis. Ann Trop Med Parasitol. 2006; 100: 715–731. https://doi.org/10.1179/ 136485906X112194 PMID: 17227650
- Zouré HGM, Wanji S, Noma M, Amazigo UV, Diggle PJ, Tekle AH, et al. The geographic distribution of Loa loa in Africa: results of large-scale implementation of the Rapid Assessment Procedure for Loiasis (RAPLOA). PLoS Negl Trop Dis. 2011; 5: e1210. https://doi.org/10.1371/journal.pntd.0001210 PMID: 21738809
- Akue JP, Nkoghe D, Padilla C, Moussavou G, Moukana H, Mbou RA, et al. Epidemiology of concomitant infection due to Loa loa and Mansonella perstans in Gabon. PLOS Neglected Tropical Diseases. 2011; 5: e1329. https://doi.org/10.1371/journal.pntd.0001329 PMID: 22022623
- Dupont A, Zue-N'dong J, Pinder M. Common occurrence of amicrofilaraemic Loa loa filariasis within the endemic region. Trans R Soc Trop Med Hyg. 1988; 82: 730. <u>https://doi.org/10.1016/0035-9203(88)</u> 90218-0 PMID: 3252591
- Pion SDS, Demanou M, Oudin B, Boussinesq M. Loiasis: the individual factors associated with the presence of microfilaraemia. Ann Trop Med Parasitol. 2005; 99: 491–500. https://doi.org/10.1179/ 136485905X51300 PMID: 16004708
- Eyebe S, Sabbagh A, Pion SD, Nana-Djeunga HC, Kamgno J, Boussinesq M, et al. Familial Aggregation and Heritability of Loa loa Microfilaremia. Clinical Infectious Diseases. 2018; 66: 751–757. https://doi.org/10.1093/cid/cix877 PMID: 29040446
- Buell KG, Whittaker C, Chesnais CB, Jewell PD, Pion SDS, Walker M, et al. Atypical clinical manifestations of loiasis and their relevance for endemic populations. Open Forum Infect Dis. 2019;6. <u>https://doi.org/10.1093/ofid/ofz417</u> PMID: 31696139
- Herrick JA, Metenou S, Makiya MA, Taylar-Williams CA, Law MA, Klion AD, et al. Eosinophil-associated processes underlie differences in clinical presentation of loiasis between temporary residents and those indigenous to Loa-endemic areas. Clin Infect Dis. 2015; 60: 55–63. https://doi.org/10.1093/cid/ciu723 PMID: 25234520
- Herrick JA, Makiya MA, Holland-Thomas N, Klion AD, Nutman TB. Infection-associated Immune Perturbations Resolve One Year Following Treatment for Loa Ioa. Clin Infect Dis. 2020. <u>https://doi.org/10.1093/cid/ciaa137</u> PMID: 32055862
- Klion AD, Massougbodji A, Sadeler BC, Ottesen EA, Nutman TB. Loiasis in endemic and nonendemic populations: immunologically mediated differences in clinical presentation. J Infect Dis. 1991; 163: 1318–1325. https://doi.org/10.1093/infdis/163.6.1318 PMID: 2037798
- Nutman TB, Miller KD, Mulligan M, Ottesen EA. Loa loa infection in temporary residents of endemic regions: recognition of a hyperresponsive syndrome with characteristic clinical manifestations. J Infect Dis. 1986; 154: 10–18. https://doi.org/10.1093/infdis/154.1.10 PMID: 3458832

- Takougang I, Meli J, Lamlenn S, Tatah PN, Ntep M. Loiasis—a neglected and under-estimated affliction: endemicity, morbidity and perceptions in eastern Cameroon. Ann Trop Med Parasitol. 2007; 101: 151–160. https://doi.org/10.1179/136485907X154511 PMID: 17316501
- Carme B, Mamboueni JP, Copin N, Noireau F. Clinical and biological study of Loa loa filariasis in Congolese. Am J Trop Med Hyg. 1989; 41: 331–337. PMID: 2679158
- 14. Pinder M. Loa loa-a neglected filaria. Parasitol Today (Regul Ed). 1988; 4: 279-284.
- Bouyou Akotet MK, Owono-Medang M, Mawili-Mboumba DP, Moussavou-Boussougou MN, Nzenze Afène S, Kendjo E, et al. The relationship between microfilaraemic and amicrofilaraemic loiasis involving co-infection with Mansonella perstans and clinical symptoms in an exposed population from Gabon. J Helminthol. 2016; 90: 469–475. https://doi.org/10.1017/S0022149X15000607 PMID: 26268068
- Mischlinger J, Veletzky L, Tazemda-Kuitsouc GB, Pitzinger P, Matsegui PB, Gmeiner M, et al. Behavioural and clinical predictors for Loiasis. J Glob Health. 2018; 8: 010413. https://doi.org/10.7189/jogh. 08.010413 PMID: 29497506
- Veletzky L, Mombo-Ngoma G, Ramharter M. The Supposedly Innocuous Eye Worm: Or, What Could Make You Put Chili in Your Eye? Am J Trop Med Hyg. 2021; 105: 1448–1449. https://doi.org/10.4269/ ajtmh.21-0319 PMID: 34516401
- Veletzky L, Hergeth J, Stelzl DR, Mischlinger J, Manego RZ, Mombo-Ngoma G, et al. Burden of disease in Gabon caused by loiasis: a cross-sectional survey. Lancet Infect Dis. 2020. <u>https://doi.org/10.1016/ S1473-3099(20)30256-5 PMID: 32585133</u>
- Chesnais CB, Takougang I, Paguélé M, Pion SD, Boussinesq M. Excess mortality associated with loiasis: a retrospective population-based cohort study. Lancet Infect Dis. 2017; 17: 108–116. <u>https://doi.org/10.1016/S1473-3099(16)30405-4 PMID: 27777031</u>
- Sandri TL, Kreidenweiss A, Cavallo S, Weber D, Juhas S, Rodi M, et al. Molecular epidemiology of Mansonella species in Gabon. J Infect Dis. 2020. <u>https://doi.org/10.1093/infdis/jiaa670</u> PMID: 33099649
- Orihel TC, editor. Bench Aids for the diagnosis of filarial infections. Geneva: World Health Organization; 1997.
- Winkler S, Willheim M, Baier K, Aichelburg A, Kremsner PG, Graninger W. Increased frequency of Th2type cytokine-producing T cells in microfilaremic loiasis. Am J Trop Med Hyg. 1999; 60: 680–686. https://doi.org/10.4269/ajtmh.1999.60.680 PMID: 10348248
- Metzger WG, Mordmüller B. Loa loa-does it deserve to be neglected? Lancet Infect Dis. 2014; 14: 353– 357. https://doi.org/10.1016/S1473-3099(13)70263-9 PMID: 24332895
- Pion SD, Chesnais CB. Uncovering the burden of loiasis: first estimates from Gabon. The Lancet Infectious Diseases. 2020;0. https://doi.org/10.1016/S1473-3099(20)30324-8 PMID: 32585134
- Jacobsen KH, Andress BC, Bhagwat EA, Bryant CA, Chandrapu VR, Desmonts CG, et al. A call for loiasis to be added to the WHO list of neglected tropical diseases. The Lancet Infectious Diseases. 2022;0. https://doi.org/10.1016/S1473-3099(22)00064-0
- Akue JP. Loa loa Pathogenesis in Humans. Human Emerging and Re-emerging Infections. John Wiley & Sons, Ltd; 2015. pp. 441–452. https://doi.org/10.1002/9781118644843.ch23
- Whittaker C, Walker M, Pion SDS, Chesnais CB, Boussinesq M, Basáñez M-G. The population biology and transmission dynamics of Loa Ioa. Trends Parasitol. 2018; 34: 335–350. https://doi.org/10.1016/j. pt.2017.12.003 PMID: 29331268
- Bhalla D, Dumas M, Preux P-M. Neurological manifestations of filarial infections. Handb Clin Neurol. 2013; 114: 235–242. https://doi.org/10.1016/B978-0-444-53490-3.00018-2 PMID: 23829914
- Gobbi F, Boussinesq M, Mascarello M, Angheben A, Gobbo M, Rossanese A, et al. Loiasis with Peripheral Nerve Involvement and Spleen Lesions. The American Journal of Tropical Medicine and Hygiene. 2011; 84: 733–737. https://doi.org/10.4269/ajtmh.2011.10–0458
- Klion AD, Nutman TB. The role of eosinophils in host defense against helminth parasites. J Allergy Clin Immunol. 2004; 113: 30–37. https://doi.org/10.1016/j.jaci.2003.10.050 PMID: 14713904
- Babu S, Nutman TB. Immune responses to helminth infection. In: Rich RR, Fleisher TA, Shearer WT, Schroeder HW, Frew AJ, Weyand CM, editors. Clinical Immunology (Fifth Edition). London: Content Repository Only!; 2019. pp. 437–447.e1. https://doi.org/10.1016/B978-0-7020-6896-6.00031–4
- Andy JJ, Ogunowo PO, Akpan NA, Odigwe CO, Ekanem IA, Esin RA. Helminth associated hypereosinophilia and tropical endomyocardial fibrosis (EMF) in Nigeria. Acta Trop. 1998; 69: 127–140. <u>https:// doi.org/10.1016/s0001-706x(97)00125-3 PMID: 9588233</u>
- **33.** Wammes LJ, Hamid F, Wiria AE, Wibowo H, Sartono E, Maizels RM, et al. Regulatory T cells in human lymphatic filariasis: stronger functional activity in microfilaremics. PLoS Negl Trop Dis. 2012; 6: e1655. https://doi.org/10.1371/journal.pntd.0001655 PMID: 22666510

- Adjobimey T, Hoerauf A. Induction of immunoglobulin G4 in human filariasis: an indicator of immunoregulation. Ann Trop Med Parasitol. 2010; 104: 455–464. https://doi.org/10.1179/ 136485910X12786389891407 PMID: 20863434
- **35.** Allen JE, Adjei O, Bain O, Hoerauf A, Hoffmann WH, Makepeace BL, et al. Of mice, cattle, and humans: the immunology and treatment of river blindness. PLoS Negl Trop Dis. 2008; 2: e217. <u>https://doi.org/10.1371/journal.pntd.0000217 PMID</u>: 18446236
- Klion AD, Law MA, Riemenschneider W, McMaster ML, Brown MR, Horne M, et al. Familial eosinophilia: a benign disorder? Blood. 2004; 103: 4050–4055. https://doi.org/10.1182/blood-2003-11-3850 PMID: 14988154
- Tamarozzi F, Buonfrate D, Ricaboni D, Ursini T, Foti G, Gobbi F. Spleen nodules in Loa loa infection: re-emerging knowledge and future perspectives. The Lancet Infectious Diseases. 2022;0. <u>https://doi.org/10.1016/S1473-3099(21)00632-0 PMID: 35219405</u>
- Akue JP, Eyang-Assengone E-R, Dieki R. Loa loa infection detection using biomarkers: current perspectives. Res Rep Trop Med. 2018; 9: 43–48. https://doi.org/10.2147/RRTM.S132380 PMID: 30050354
- Fink DL, Kamgno J, Nutman TB. Rapid molecular assays for specific detection and quantitation of Loa loa microfilaremia. PLoS Negl Trop Dis. 2011; 5: e1299. https://doi.org/10.1371/journal.pntd.0001299 PMID: 21912716
- Akue JP, Devaney E, Wahl G, Moukana H. Expression of filarial-specific IgG subclasses under different transmission intensities in a region endemic for loiasis. Am J Trop Med Hyg. 2002; 66: 245–250. <u>https:// doi.org/10.4269/ajtmh.2002.66.245</u> PMID: 12139215
- 41. Van Hoegaerden M, Chabaud B, Akue JP, Ivanoff B. Filariasis due to Loa loa and Mansonella perstans: distribution in the region of Okondja, Haut-Ogooué Province, Gabon, with parasitological and serological follow-up over one year. Trans R Soc Trop Med Hyg. 1987; 81: 441–446. <u>https://doi.org/10.1016/</u> 0035-9203(87)90163-5 PMID: 3479854
- Takougang I, Meremikwu M, Wandji S, Yenshu EV, Aripko B, Lamlenn SB, et al. Rapid assessment method for prevalence and intensity of Loa loa infection. Bull World Health Organ. 2002; 80: 852–858. PMID: 12481206